Author’s response to reviews

Title: Targeted individual exercise programmes for older medical patients are feasible, and changes hospital and patient outcomes: a service improvement project

Authors:

Jo S Nolan (jo.nolan@fmc.sa.gov.au)
Susie K Thomas (thom0767@flinders.edu.au)

Version: 7 Date: 3 December 2008

Author’s response to reviews:

3rd December 2008

Dear Dr Alexandersson,

Please find a revised version of the manuscript and figures, addressing all of the formatting issues that were highlighted. One of the revisions included a comment that Reference 13 was not cited in the text, however, it can be found on page 15, paragraph 3, line 4 of the manuscript.

I hope the manuscript now meets all of your requirements.

Kind Regards,

Susie Thomas